Author's response to reviews

Title: Household headship and child death: Evidence from Nepal

Authors:

Ramesh Adhikari (rameships@gmail.com)
Chai Podhisita (prcps@mahidol.ac.th)

Version: 4 Date: 23 May 2010

Author's response to reviews: see over
Dear Editor,

We have incorporated comment given by the reviewer. Attached please find a copy of the revised version of our manuscript entitled "Household headship and child death: Evidence from Nepal"

Regards,
Ramesh Adhikari

Comment by Margaret Elizabeth Kruk:

I have one further point of clarification: the final adjusted regression coefficients do not appear to have changed as a result of adding a new variable (birth interval)--this is based on what the authors indicated in track changes. When a new variable is added, we would expect the other coefficients to change (particularly as the new variable is highly significant). The authors need to double-check their model and update the coefficients.

Response: Thank you very much for the comment. We have checked the model and updated the coefficients. Correction is in track changed format in our revised manuscript.

Once again, thank you so much for your comment.